Author's response to reviews

Title: Particulate matter components and subclinical atherosclerosis: common approaches to estimating exposure in a Multi-Ethnic Study of Atherosclerosis cross-sectional study

Authors:

Sun Min (min0607_sun@hotmail.com)
Joel D Kaufman (joelk@u.washington.edu)
Kim Sun-Young (puha0@u.washington.edu)
Timothy Larson (tlarson@u.washington.edu)
Timothy Gould (tgould@u.washington.edu)
Joseph F Polak (jpolak@tuftsmedicalcenter.org)
Matthew J Budoff (mbudoff@labiomed.org)
Ana V Diez Roux (adiezrou@umich.edu)
Sverre Vedal (svedal@u.washington.edu)

Version: 4 Date: 7 April 2013

Author's response to reviews:

A sentence was added to the first paragraph of the Methods confirming IRB approval of study and participant consent.

"Institutional review board approval was granted at each study site and written informed consent was obtained from all participants."